Author's response to reviews

Title: Risk of myocardial infarction in parents of HIV-infected individuals: a population-based cohort study.

Authors:

Line D Rasmussen (linedahlerup@hotmail.com)
Lars H Omland (omland@dadlenet.dk)
Court Pedersen (court.pedersen@ouh.regionsyddanmark.dk)
Jan Gerstoft (jan.gerstoft@rh.regionh.dk)
Gitte Kronborg (gkronborg@dadlnet.dk)
Janne Jensen (janne.jensen@fks.regionsyddanmark.dk)
Niels Obel (niels.obel@rh.regionh.dk)

Version: 7 Date: 9 June 2010

Author's response to reviews: see over
To the Editor, BMC Infectious Disease.

We have now changed the format of the manuscript “Risk of Myocardial Infarction of HIV-infected Individuals: a population based cohort Study” according to the additional revisions listed in the mail of June 9, 2010.

Major revisions:

1) An authors’ contribution section has been including directly preceding the acknowledgments. The format of the authors’ contribution previously attached to the journal has been used, but the names have been changed to initials. The names in the conflicts of interest section have also been changed to initials.

2) Highlights and tracking have all been removed.

Yours sincerely

Line Dahlerup Rasmussen.

Corresponding author:

Line Dahlerup Rasmussen
Department of Infectious Disease
Odense University Hospital
Sønderboulevard 29
5000 Odense C.
Denmark

Email: linedahlerup@hotmail.com
Telephone: +45 65411321
Fax: +45 66117418